There are 2 reviews. First review did not ask for any revision. Second review is as follows, "I consider this manuscript as a very useful review of iris expantion devises.

The only minor concern is that the refferneces should be updated because are too old in my opinion." We have already answered and revised the manuscript according to this reviewer's suggestion. We had increased the references from 11 to 26 and provided several more recent references as well.